Author’s response to reviews

Title: Amyand's hernia: a case series

Authors:

Kyriakos Psarras (psarras_k@yahoo.com)
Miltiadis A Lalountas (miltiadiaslalountas@yahoo.gr)
Minas Baltatzis (minbalt@yahoo.gr)
Efstathios Pavlidis (pavlidakos@hotmail.com)
Anastasios Tsitlakidis (skrits7@hotmail.com)
Nikolaos Symeonidis (niksym@hotmail.com)
Konstantinos Ballas (mpallas@auth.gr)
Theodoros Pavlidis (pavlidth@med.auth.gr)
Athanassios Sakantamis (asakantam@med.auth.gr)

Version: 8 Date: 25 July 2011

Author’s response to reviews: see over
Reply to the Editors’ comments

With respect to the Editors’ comments, we made all the necessary changes in this revised version of the manuscript.

1. In the introduction section of the abstract, the sentence “The finding of vermiform appendix in inguinal hernia, inflamed or not, is called Amyand's hernia.” changed to “The finding of a vermiform appendix in an inguinal hernia, inflamed or not, is called Amyand's hernia.”

2. In the case presentations section of the abstract, the sentence “One of them had an inflamed vermiform appendix in his sac (acute appendicitis), presented as incarcerated right groin hernia and underwent simultaneous appendicectomy and Bassini’s suture hernia repair.” replaced with “One of them had an inflamed vermiform appendix in his sac (acute appendicitis), presenting as an incarcerated right groin hernia and underwent simultaneous appendectomy and Bassini’s suture hernia repair.”

3. In the introduction section of the main text, the following sentence “Claudius Amyand (1660–1740) a French surgeon working at St. George’s and Westminster hospitals in London, performed the first successful appendectomy in an 11-year-old boy presented with an inflamed, perforated appendix in an inguinal hernia sac in 1735.” replaced with “Claudius Amyand (1660–1740) a French surgeon working at St. George’s and Westminster hospitals in London, performed the first successful appendectomy in an 11-year-old boy who presented with an inflamed, perforated appendix in an inguinal hernia sac in 1735.”


The authors would like to thank the Editors for their cooperation.

Miltiadis Lalountas,

Surgeon MD, MSc, PhD Candidate